Author’s response to reviews

Title: A randomized, double-blind, positive-controlled, prospective, dose-response clinical study to evaluate the efficacy and tolerability of an aqueous extract of Terminalia bellerica in lowering uric acid and creatinine levels in chronic kidney disease subjects with hyperuricemia.

Authors:

Usharani Pingali (ushapingali@yahoo.com)
Chandrakekhar Nutralapati (csnpru@gmail.com)
Niranjan Koilagundla (dr.niranjandm@gmail.com)
Gangadhar Taduri (gangadhar3@yahoo.co.in)

Version: 4 Date: 01 Sep 2020

Author’s response to reviews:

Reply to the reviewers comments has been uploaded as word document